Author’s response to reviews

Title: Perioperative management of paraganglioma and catecholamine-induced cardiomyopathy in child- a case report and review of the literature

Authors:

Xixi Jia (13811452657@163.com)
Xiangyang Guo (xiangyangguoanesth@126.com)
Qing Zheng (cici-user@sohu.com)

Version: 2 Date: 23 Sep 2017

Author’s response to reviews:

To honorable editor and reviewers:

Thank you for your pertinent comments on my article. I have revised the article and made the necessary corrections according with your comments:

1. Re-phrased several sections in my own words to avoid direct overlap.
2. The Consent for Publication is separated from Consent to Participate.
3. Represent authors' names using full initials in the Authors’ Contributions section.
4. Add a section 'Figure legends' listing all the figure legends after the reference.
5. I ensure that it is in the final form for publication.